\providecommand{\href}[2]{#2}\begingroup\raggedright\begin{thebibliography}{10}

\bibitem{Chatrchyan:2012ufa}
\textbf{CMS} Collaboration, S.~Chatrchyan et~al., {\it {Observation of a new
  boson at a mass of 125 GeV with the CMS experiment at the LHC}},  {\em
  Phys.Lett.} {\bf B716} (2012) 30--61,
  [\href{http://xxx.lanl.gov/abs/1207.7235}{{\tt arXiv:1207.7235}}].

\bibitem{Aad:2012tfa}
\textbf{ATLAS} Collaboration, G.~Aad et~al., {\it {Observation of a new
  particle in the search for the Standard Model Higgs boson with the ATLAS
  detector at the LHC}},  {\em Phys.Lett.} {\bf B716} (2012) 1--29,
  [\href{http://xxx.lanl.gov/abs/1207.7214}{{\tt arXiv:1207.7214}}].

\bibitem{Djouadi:2012ae}
A.~Djouadi and A.~Lenz, {\it {Sealing the fate of a fourth generation of
  fermions}},  {\em Phys.Lett.} {\bf B715} (2012) 310--314,
  [\href{http://xxx.lanl.gov/abs/1204.1252}{{\tt arXiv:1204.1252}}].

\bibitem{Kuflik:2012ai}
E.~Kuflik, Y.~Nir, and T.~Volansky, {\it {Implications of Higgs Searches on the
  Four Generation Standard Model}},
  \href{http://xxx.lanl.gov/abs/1204.1975}{{\tt arXiv:1204.1975}}.

\bibitem{Eberhardt:2012sb}
O.~Eberhardt, G.~Herbert, H.~Lacker, A.~Lenz, A.~Menzel, U.~Nierste, and
  M.~Wiebusch, {\it {Joint analysis of Higgs decays and electroweak precision
  observables in the Standard Model with a sequential fourth generation}},
  {\em Phys.Rev.} {\bf D86} (2012) 013011,
  [\href{http://xxx.lanl.gov/abs/1204.3872}{{\tt arXiv:1204.3872}}].

\bibitem{Buchkremer:2012yy}
M.~Buchkremer, J.-M. Gerard, and F.~Maltoni, {\it {Closing in on a perturbative
  fourth generation}},  {\em JHEP} {\bf 1206} (2012) 135,
  [\href{http://xxx.lanl.gov/abs/1204.5403}{{\tt arXiv:1204.5403}}].

\bibitem{Eberhardt:2012ck}
O.~Eberhardt, A.~Lenz, A.~Menzel, U.~Nierste, and M.~Wiebusch, {\it {Status of
  the fourth fermion generation before ICHEP2012: Higgs data and electroweak
  precision observables}},  \href{http://xxx.lanl.gov/abs/1207.0438}{{\tt
  arXiv:1207.0438}}.

\bibitem{Eberhardt:2012gv}
O.~Eberhardt, G.~Herbert, H.~Lacker, A.~Lenz, A.~Menzel, U.~Nierste, and
  M.~Wiebusch, {\it {Impact of a Higgs boson at a mass of 126 GeV on the
  standard model with three and four fermion generations}},  {\em
  Phys.Rev.Lett.} {\bf 109} (2012) 241802,
  [\href{http://xxx.lanl.gov/abs/1209.1101}{{\tt arXiv:1209.1101}}].

\bibitem{Lee:1973iz}
T.~Lee, {\it {A Theory of Spontaneous T Violation}},  {\em Phys.Rev.} {\bf D8}
  (1973) 1226--1239.

\bibitem{Glashow:1976nt}
S.~L. Glashow and S.~Weinberg, {\it {Natural Conservation Laws for Neutral
  Currents}},  {\em Phys.Rev.} {\bf D15} (1977) 1958.

\bibitem{Donoghue:1978cj}
J.~F. Donoghue and L.~F. Li, {\it {Properties of Charged Higgs Bosons}},  {\em
  Phys.Rev.} {\bf D19} (1979) 945.

\bibitem{Deschamps:2009rh}
O.~Deschamps, S.~Descotes-Genon, S.~Monteil, V.~Niess, S.~T'Jampens, et~al.,
  {\it {The Two Higgs Doublet of Type II facing flavour physics data}},  {\em
  Phys.Rev.} {\bf D82} (2010) 073012,
  [\href{http://xxx.lanl.gov/abs/0907.5135}{{\tt arXiv:0907.5135}}].

\bibitem{Carmi:2012in}
D.~Carmi, A.~Falkowski, E.~Kuflik, T.~Volansky, and J.~Zupan, {\it {Higgs After
  the Discovery: A Status Report}},
  \href{http://xxx.lanl.gov/abs/1207.1718}{{\tt arXiv:1207.1718}}.

\bibitem{Cheon:2012rh}
H.~Cheon and S.~K. Kang, {\it {Constraining parameter space in type-II
  two-Higgs doublet model in light of a 125 GeV Higgs boson}},
  \href{http://xxx.lanl.gov/abs/1207.1083}{{\tt arXiv:1207.1083}}.

\bibitem{Drozd:2012vf}
A.~Drozd, B.~Grzadkowski, J.~F. Gunion, and Y.~Jiang, {\it {Two-Higgs-Doublet
  Models and Enhanced Rates for a 125 GeV Higgs}},
  \href{http://xxx.lanl.gov/abs/1211.3580}{{\tt arXiv:1211.3580}}.

\bibitem{Ferreira:2011aa}
P.~Ferreira, R.~Santos, M.~Sher, and J.~P. Silva, {\it {Implications of the LHC
  two-photon signal for two-Higgs-doublet models}},
  \href{http://xxx.lanl.gov/abs/1112.3277}{{\tt arXiv:1112.3277}}.

\bibitem{Chen:2013kt}
C.-Y. Chen and S.~Dawson, {\it {Exploring Two Higgs Doublet Models Through
  Higgs Production}},  \href{http://xxx.lanl.gov/abs/1301.0309}{{\tt
  arXiv:1301.0309}}.

\bibitem{Celis:2013rcs}
A.~Celis, V.~Ilisie, and A.~Pich, {\it {LHC constraints on two-Higgs doublet
  models}},  \href{http://xxx.lanl.gov/abs/1302.4022}{{\tt arXiv:1302.4022}}.

\bibitem{Giardino:2013bma}
P.~P. Giardino, K.~Kannike, I.~Masina, M.~Raidal, and A.~Strumia, {\it {The
  universal Higgs fit}},  \href{http://xxx.lanl.gov/abs/1303.3570}{{\tt
  arXiv:1303.3570}}.

\bibitem{Grinstein:2013npa}
B.~Grinstein and P.~Uttayarat, {\it {Carving Out Parameter Space in Type-II Two
  Higgs Doublets Model}},  \href{http://xxx.lanl.gov/abs/1304.0028}{{\tt
  arXiv:1304.0028}}.

\bibitem{Barroso:2013zxa}
A.~Barroso, P.~Ferreira, R.~Santos, M.~Sher, and J.~P. Silva, {\it {2HDM at the
  LHC - the story so far}},  \href{http://xxx.lanl.gov/abs/1304.5225}{{\tt
  arXiv:1304.5225}}.

\bibitem{Coleppa:2013dya}
B.~Coleppa, F.~Kling, and S.~Su, {\it {Constraining Type II 2HDM in Light of
  LHC Higgs Searches}},  \href{http://xxx.lanl.gov/abs/1305.0002}{{\tt
  arXiv:1305.0002}}.

\bibitem{Gunion:2002zf}
J.~F. Gunion and H.~E. Haber, {\it {The CP conserving two Higgs doublet model:
  The Approach to the decoupling limit}},  {\em Phys.Rev.} {\bf D67} (2003)
  075019, [\href{http://xxx.lanl.gov/abs/hep-ph/0207010}{{\tt
  hep-ph/0207010}}].

\bibitem{Barroso:2013awa}
A.~Barroso, P.~Ferreira, I.~Ivanov, and R.~Santos, {\it {Metastability bounds
  on the two Higgs doublet model}},
  \href{http://xxx.lanl.gov/abs/1303.5098}{{\tt arXiv:1303.5098}}.

\bibitem{Nierste:1995zx}
U.~Nierste and K.~Riesselmann, {\it {Higgs sector renormalization group in the
  MS and OMS scheme: The Breakdown of perturbation theory for a heavy Higgs}},
  {\em Phys.Rev.} {\bf D53} (1996) 6638--6652,
  [\href{http://xxx.lanl.gov/abs/hep-ph/9511407}{{\tt hep-ph/9511407}}].

\bibitem{ATLAS-CONF-2013-012}
{ATLAS conference note, ATLAS-CONF-2013-012}.

\bibitem{ATLAS-CONF-2013-013}
{ATLAS conference note, ATLAS-CONF-2013-013}.

\bibitem{CMS-PAS-HIG-13-001}
{CMS physics analysis summary, CMS-PAS-HIG-13-001}.

\bibitem{CMS-PAS-HIG-13-002}
{CMS physics analysis summary, CMS-PAS-HIG-13-002}.

\bibitem{Ferreira:2012my}
P.~Ferreira, R.~Santos, M.~Sher, and J.~P. Silva, {\it {Could the LHC
  two-photon signal correspond to the heavier scalar in two-Higgs-doublet
  models?}},  {\em Phys.Rev.} {\bf D85} (2012) 035020,
  [\href{http://xxx.lanl.gov/abs/1201.0019}{{\tt arXiv:1201.0019}}].

\bibitem{Burdman:2011ki}
G.~Burdman, C.~E. Haluch, and R.~D. Matheus, {\it {Is the LHC Observing the
  Pseudo-scalar State of a Two-Higgs Doublet Model ?}},  {\em Phys.Rev.} {\bf
  D85} (2012) 095016, [\href{http://xxx.lanl.gov/abs/1112.3961}{{\tt
  arXiv:1112.3961}}].

\bibitem{Ferreira:2012nv}
P.~Ferreira, H.~E. Haber, R.~Santos, and J.~P. Silva, {\it {Mass-degenerate
  Higgs bosons at 125 GeV in the Two-Higgs-Doublet Model}},
  \href{http://xxx.lanl.gov/abs/1211.3131}{{\tt arXiv:1211.3131}}.

\bibitem{Hahn:1998yk}
T.~Hahn and M.~Perez-Victoria, {\it {Automatized one-loop calculations in four
  and D dimensions}},  {\em Comput. Phys. Commun.} {\bf 118} (1999) 153--165,
  [\href{http://xxx.lanl.gov/abs/hep-ph/9807565}{{\tt hep-ph/9807565}}].

\bibitem{Hahn:2000kx}
T.~Hahn, {\it {Generating Feynman diagrams and amplitudes with FeynArts 3}},
  {\em Comput. Phys. Commun.} {\bf 140} (2001) 418--431,
  [\href{http://xxx.lanl.gov/abs/hep-ph/0012260}{{\tt hep-ph/0012260}}].

\bibitem{Hahn:2006qw}
T.~Hahn and M.~Rauch, {\it {News from FormCalc and LoopTools}},  {\em Nucl.
  Phys. Proc. Suppl.} {\bf 157} (2006) 236--240,
  [\href{http://xxx.lanl.gov/abs/hep-ph/0601248}{{\tt hep-ph/0601248}}].

\bibitem{CMS-PAS-HIG-13-003}
{CMS physics analysis summary, CMS-PAS-HIG-13-003}.

\bibitem{Hollik:1986gg}
W.~Hollik, {\it {Nonstandard Higgs bosons in $SU(2)\times U(1)$ radiative
  corrections}},  {\em Z.Phys.} {\bf C32} (1986) 291.

\bibitem{Hollik:1987fg}
W.~Hollik, {\it {Radiative corrections with two Higgs doublets at LEP/SLC and
  HERA}},  {\em Z.Phys.} {\bf C37} (1988) 569.

\bibitem{Haber:1999zh}
H.~E. Haber and H.~E. Logan, {\it {Radiative corrections to the Z b anti-b
  vertex and constraints on extended Higgs sectors}},  {\em Phys.Rev.} {\bf
  D62} (2000) 015011, [\href{http://xxx.lanl.gov/abs/hep-ph/9909335}{{\tt
  hep-ph/9909335}}].

\bibitem{Gonzalez:2011he}
P.~Gonzalez, J.~Rohrwild, and M.~Wiebusch, {\it {Electroweak Precision
  Observables within a Fourth Generation Model with General Flavour
  Structure}},  {\em Eur.Phys.J.} {\bf C72} (2012) 2007,
  [\href{http://xxx.lanl.gov/abs/1105.3434}{{\tt arXiv:1105.3434}}].

\bibitem{Bardin:1989tq}
D.~Y. Bardin, M.~S. Bilenky, T.~Riemann, M.~Sachwitz, and H.~Vogt, {\it {DIZET:
  A Program Package for the Calculation of Electroweak One Loop Corrections for
  the Process $e^+e^-\to f^+f^-$ Around the $Z^0$ Peak}},  {\em
  Comput.Phys.Commun.} {\bf 59} (1990) 303--312.

\bibitem{Bardin:1999yd}
D.~Y. Bardin, P.~Christova, M.~Jack, L.~Kalinovskaya, A.~Olchevski, et~al.,
  {\it {ZFITTER v.6.21: A Semianalytical program for fermion pair production in
  e+ e- annihilation}},  {\em Comput.Phys.Commun.} {\bf 133} (2001) 229--395,
  [\href{http://xxx.lanl.gov/abs/hep-ph/9908433}{{\tt hep-ph/9908433}}].

\bibitem{Arbuzov:2005ma}
A.~Arbuzov, M.~Awramik, M.~Czakon, A.~Freitas, M.~Grunewald, et~al., {\it
  {ZFITTER: A Semi-analytical program for fermion pair production in $e^+e^-$
  annihilation, from version 6.21 to version 6.42}},  {\em Comput.Phys.Commun.}
  {\bf 174} (2006) 728--758,
  [\href{http://xxx.lanl.gov/abs/hep-ph/0507146}{{\tt hep-ph/0507146}}].

\bibitem{Freitas:2012sy}
A.~Freitas and Y.-C. Huang, {\it {Electroweak two-loop corrections to
  $sin^2{\theta_{\text{eff},bb}}$ and $R_b$ using numerical Mellin-Barnes
  integrals}},  {\em JHEP} {\bf 1208} (2012) 050,
  [\href{http://xxx.lanl.gov/abs/1205.0299}{{\tt arXiv:1205.0299}}].

\bibitem{Ciuchini:1997xe}
M.~Ciuchini, G.~Degrassi, P.~Gambino, and G.~Giudice, {\it {Next-to-leading QCD
  corrections to $B\to X_s\gamma$: Standard model and two Higgs doublet
  model}},  {\em Nucl.Phys.} {\bf B527} (1998) 21--43,
  [\href{http://xxx.lanl.gov/abs/hep-ph/9710335}{{\tt hep-ph/9710335}}].

\bibitem{Borzumati:1998tg}
F.~Borzumati and C.~Greub, {\it {2HDMs predictions for $\bar B\to X_s\gamma$ in
  NLO QCD}},  {\em Phys.Rev.} {\bf D58} (1998) 074004,
  [\href{http://xxx.lanl.gov/abs/hep-ph/9802391}{{\tt hep-ph/9802391}}].

\bibitem{Borzumati:1998nx}
F.~Borzumati and C.~Greub, {\it {Two Higgs doublet model predictions for $\bar
  B\to X_s\gamma$ in NLO QCD: Addendum}},  {\em Phys.Rev.} {\bf D59} (1999)
  057501, [\href{http://xxx.lanl.gov/abs/hep-ph/9809438}{{\tt
  hep-ph/9809438}}].

\bibitem{Ciafaloni:1997un}
P.~Ciafaloni, A.~Romanino, and A.~Strumia, {\it {Two loop QCD corrections to
  charged Higgs mediated $b\to s\gamma$ decay}},  {\em Nucl.Phys.} {\bf B524}
  (1998) 361--376, [\href{http://xxx.lanl.gov/abs/hep-ph/9710312}{{\tt
  hep-ph/9710312}}].

\bibitem{Bobeth:1999ww}
C.~Bobeth, M.~Misiak, and J.~Urban, {\it {Matching conditions for $b\to
  s\gamma$ and $b\to sg$ in extensions of the standard model}},  {\em
  Nucl.Phys.} {\bf B567} (2000) 153--185,
  [\href{http://xxx.lanl.gov/abs/hep-ph/9904413}{{\tt hep-ph/9904413}}].

\bibitem{Hermann:2012fc}
T.~Hermann, M.~Misiak, and M.~Steinhauser, {\it {$\bar{B}\to X_s \gamma$ in the
  Two Higgs Doublet Model up to Next-to-Next-to-Leading Order in QCD}},  {\em
  JHEP} {\bf 1211} (2012) 036, [\href{http://xxx.lanl.gov/abs/1208.2788}{{\tt
  arXiv:1208.2788}}].

\bibitem{Misiak:2006ab}
M.~Misiak and M.~Steinhauser, {\it {NNLO QCD corrections to the $\bar B \to
  X_s\gamma$ matrix elements using interpolation in $m_c$}},  {\em Nucl.Phys.}
  {\bf B764} (2007) 62--82, [\href{http://xxx.lanl.gov/abs/hep-ph/0609241}{{\tt
  hep-ph/0609241}}].

\bibitem{Amhis:2012bh}
\textbf{Heavy Flavor Averaging Group}, Y.~Amhis et~al., {\it {Averages of
  B-Hadron, C-Hadron, and tau-lepton properties as of early 2012}},
  \href{http://xxx.lanl.gov/abs/1207.1158}{{\tt arXiv:1207.1158}}.

\bibitem{Gambino:2008fj}
P.~Gambino and P.~Giordano, {\it {Normalizing inclusive rare B decays}},  {\em
  Phys.Lett.} {\bf B669} (2008) 69--73,
  [\href{http://xxx.lanl.gov/abs/0805.0271}{{\tt arXiv:0805.0271}}].

\bibitem{Abbott:1979dt}
L.~Abbott, P.~Sikivie, and M.~B. Wise, {\it {Constraints on Charged Higgs
  Couplings}},  {\em Phys.Rev.} {\bf D21} (1980) 1393.

\bibitem{Geng:1988bq}
C.~Geng and J.~N. Ng, {\it {CHARGED HIGGS EFFECT IN B)d)0 - anti-B(d)0 MIXING,
  K $\to$ pi neutrino anti-neutrino DECAY AND RARE DECAYS OF B MESONS}},  {\em
  Phys.Rev.} {\bf D38} (1988) 2857.

\bibitem{Buras:1989ui}
A.~J. Buras, P.~Krawczyk, M.~E. Lautenbacher, and C.~Salazar, {\it {$B^0$-$\bar
  B^0$ mixing, $CP$ violation, $K^+\to\pi^+\nu\bar\nu$ and $B\to K\gamma X$ in
  a two Higgs doublet model}},  {\em Nucl.Phys.} {\bf B337} (1990) 284--312.

\bibitem{CKMfitter:2012mo}
{Private communication with the CKMfitter group}.

\bibitem{Buras:1990fn}
A.~J. Buras, M.~Jamin, and P.~H. Weisz, {\it {Leading and next-to-leading QCD
  corrections to $\epsilon$ parameter and $B^0$-$\bar B^0$ mixing in the
  presence of a heavy top quark}},  {\em Nucl.Phys.} {\bf B347} (1990)
  491--536.

\bibitem{Beringer:1900zz}
\textbf{Particle Data Group}, J.~Beringer et~al., {\it {Review of Particle
  Physics (RPP)}},  {\em Phys.Rev.} {\bf D86} (2012) 010001.

\bibitem{Lenz:2012az}
A.~Lenz, U.~Nierste, J.~Charles, S.~Descotes-Genon, H.~Lacker, et~al., {\it
  {Constraints on new physics in $B-\bar{B}$ mixing in the light of recent LHCb
  data}},  {\em Phys.Rev.} {\bf D86} (2012) 033008,
  [\href{http://xxx.lanl.gov/abs/1203.0238}{{\tt arXiv:1203.0238}}].

\bibitem{Gamiz:2009ku}
\textbf{HPQCD} Collaboration, E.~Gamiz, C.~T. Davies, G.~P. Lepage,
  J.~Shigemitsu, and M.~Wingate, {\it {Neutral $B$ Meson Mixing in Unquenched
  Lattice QCD}},  {\em Phys.Rev.} {\bf D80} (2009) 014503,
  [\href{http://xxx.lanl.gov/abs/0902.1815}{{\tt arXiv:0902.1815}}].

\bibitem{Aaij:2013mpa}
\textbf{LHCb}, R.~Aaij et~al., {\it {Precision measurement of the $B^0_s -
  \bar{B}^0_s$ oscillation frequency with the decay $B^0_s \to D^-_s\pi^+$}},
  \href{http://xxx.lanl.gov/abs/1304.4741}{{\tt arXiv:1304.4741}}.

\bibitem{Hou:1992sy}
W.-S. Hou, {\it {Enhanced charged Higgs boson effects in $B^-$ to tau
  anti-neutrino, mu anti-neutrino and $b$ to tau anti-neutrino + X}},  {\em
  Phys.Rev.} {\bf D48} (1993) 2342--2344.

\bibitem{Akeroyd:2003zr}
A.~Akeroyd and S.~Recksiegel, {\it {The Effect of H+- on B+- $\to$ tau+-
  nu(tau) and B+- $\to$ mu+- muon neutrino}},  {\em J.Phys.} {\bf G29} (2003)
  2311--2317, [\href{http://xxx.lanl.gov/abs/hep-ph/0306037}{{\tt
  hep-ph/0306037}}].

\bibitem{Grzadkowski:1992qj}
B.~Grzadkowski and W.-S. Hou, {\it {Searching for $B \to D \tau \bar{\tau}$-
  neutrino at the 10-percent level}},  {\em Phys.Lett.} {\bf B283} (1992)
  427--433.

\bibitem{Miki:2002nz}
T.~Miki, T.~Miura, and M.~Tanaka, {\it {Effects of charged Higgs boson and QCD
  corrections in anti-B $\to$ D tau anti-nu(tau)}},
  \href{http://xxx.lanl.gov/abs/hep-ph/0210051}{{\tt hep-ph/0210051}}.

\bibitem{Nierste:2008qe}
U.~Nierste, S.~Trine, and S.~Westhoff, {\it {Charged-Higgs effects in a new B
  $\to$ D tau nu differential decay distribution}},  {\em Phys.Rev.} {\bf D78}
  (2008) 015006, [\href{http://xxx.lanl.gov/abs/0801.4938}{{\tt
  arXiv:0801.4938}}].

\bibitem{Kamenik:2008tj}
J.~F. Kamenik and F.~Mescia, {\it {B $\to$ D tau nu Branching Ratios:
  Opportunity for Lattice QCD and Hadron Colliders}},  {\em Phys.Rev.} {\bf
  D78} (2008) 014003, [\href{http://xxx.lanl.gov/abs/0802.3790}{{\tt
  arXiv:0802.3790}}].

\bibitem{Trine:2008qv}
S.~Trine, {\it {Charged-Higgs effects in B $\to$ (D) tau nu decays}},
  \href{http://xxx.lanl.gov/abs/0810.3633}{{\tt arXiv:0810.3633}}.

\bibitem{Tanaka:2010se}
M.~Tanaka and R.~Watanabe, {\it {Tau longitudinal polarization in B $\to$ D tau
  nu and its role in the search for charged Higgs boson}},  {\em Phys.Rev.}
  {\bf D82} (2010) 034027, [\href{http://xxx.lanl.gov/abs/1005.4306}{{\tt
  arXiv:1005.4306}}].

\bibitem{Fajfer:2012vx}
S.~Fajfer, J.~F. Kamenik, and I.~Nisandzic, {\it {On the $B \to D^* \tau
  \bar\nu_\tau$ Sensitivity to New Physics}},  {\em Phys.Rev.} {\bf D85} (2012)
  094025, [\href{http://xxx.lanl.gov/abs/1203.2654}{{\tt arXiv:1203.2654}}].

\bibitem{Sakaki:2012ft}
Y.~Sakaki and H.~Tanaka, {\it {Constraints of the Charged Scalar Effects Using
  the Forward-Backward Asymmetry on $B\to D^{(*)}\tau\bar{\nu_{\tau}}$}},  {\em
  Phys.Rev.} {\bf D87} (2013) 054002,
  [\href{http://xxx.lanl.gov/abs/1205.4908}{{\tt arXiv:1205.4908}}].

\bibitem{Becirevic:2012jf}
D.~Becirevic, N.~Kosnik, and A.~Tayduganov, {\it {$\bar B\to D\tau\bar
  \nu_\tau$ vs. $\bar B\to D\mu\bar \nu_\mu$}},  {\em Phys.Lett.} {\bf B716}
  (2012) 208--213, [\href{http://xxx.lanl.gov/abs/1206.4977}{{\tt
  arXiv:1206.4977}}].

\bibitem{Celis:2012dk}
A.~Celis, M.~Jung, X.-Q. Li, and A.~Pich, {\it {Sensitivity to charged scalars
  in $B\to D^{(*)}\tau\nu_\tau$ and $B\to\tau\nu_\tau$ decays}},  {\em JHEP}
  {\bf 1301} (2013) 054, [\href{http://xxx.lanl.gov/abs/1210.8443}{{\tt
  arXiv:1210.8443}}].

\bibitem{Tanaka:2012nw}
M.~Tanaka and R.~Watanabe, {\it {New physics in the weak interaction of $\bar
  B\to D^{(*)}\tau\bar\nu$}},  {\em Phys.Rev.} {\bf D87} (2013) 034028,
  [\href{http://xxx.lanl.gov/abs/1212.1878}{{\tt arXiv:1212.1878}}].

\bibitem{Logan:2000iv}
H.~E. Logan and U.~Nierste, {\it {$B_{s, d} \to \ell^+ \ell^-$ in a two Higgs
  doublet model}},  {\em Nucl.Phys.} {\bf B586} (2000) 39--55,
  [\href{http://xxx.lanl.gov/abs/hep-ph/0004139}{{\tt hep-ph/0004139}}].

\bibitem{Crivellin:2012ye}
A.~Crivellin, C.~Greub, and A.~Kokulu, {\it {Explaining $B\to D\tau\nu$, $B\to
  D^*\tau\nu$ and $B\to \tau\nu$ in a 2HDM of type III}},  {\em Phys.Rev.} {\bf
  D86} (2012) 054014, [\href{http://xxx.lanl.gov/abs/1206.2634}{{\tt
  arXiv:1206.2634}}].

\bibitem{Wiebusch:2012en}
M.~Wiebusch, {\it {Numerical Computation of p-values with myFitter}},
  \href{http://xxx.lanl.gov/abs/1207.1446}{{\tt arXiv:1207.1446}}.

\bibitem{Hocker:2001xe}
A.~Hocker, H.~Lacker, S.~Laplace, and F.~Le~Diberder, {\it {A New approach to a
  global fit of the CKM matrix}},  {\em Eur.Phys.J.} {\bf C21} (2001) 225--259,
  [\href{http://xxx.lanl.gov/abs/hep-ph/0104062}{{\tt hep-ph/0104062}}].

\bibitem{CMS-PAS-HIG-12-043}
{CMS physics analysis summary, CMS-PAS-HIG-12-043}.

\bibitem{ATLAS-CONF-2012-091a}
{ATLAS conference note, ATLAS-CONF-2012-091, Table 6}.

\bibitem{CDF:2013kxa}
\textbf{CDF, D0} Collaborations, {\it {Higgs Boson Studies at the Tevatron}},
  \href{http://xxx.lanl.gov/abs/1303.6346}{{\tt arXiv:1303.6346}}.

\bibitem{Aad:2012an}
\textbf{ATLAS} Collaboration, G.~Aad et~al., {\it {Combined search for the
  Standard Model Higgs boson in $pp$ collisions at $\sqrt{s} = 7$ TeV with the
  ATLAS detector}},  {\em Phys.Rev.} {\bf D86} (2012) 032003,
  [\href{http://xxx.lanl.gov/abs/1207.0319}{{\tt arXiv:1207.0319}}].

\bibitem{ATLAS-CONF-2012-091}
{ATLAS conference note, ATLAS-CONF-2012-091, Fig. 14a}.

\bibitem{ATLAS-CONF-2013-030}
{ATLAS conference note, ATLAS-CONF-2013-030}.

\bibitem{ATLAS-CONF-2012-161}
{ATLAS conference note, ATLAS-CONF-2012-161}.

\bibitem{ATLAS-CONF-2012-160}
{ATLAS conference note, ATLAS-CONF-2012-160}.

\bibitem{CMS-PAS-HIG-12-020}
{CMS physics analysis summary, CMS-PAS-HIG-12-020}.

\bibitem{CMS-PAS-HIG-12-045}
{CMS physics analysis summary, CMS-PAS-HIG-12-045}.

\bibitem{CMS-PAS-HIG-13-004}
{CMS physics analysis summary, CMS-PAS-HIG-13-004}.

\end{thebibliography}\endgroup

\begin{table}[b]
{\centering\small
\begin{tabular}{llllll}
\toprule
             & ggF & VBF & VH \\
\midrule
0/1 jet [\%] & $0.945$ & $2.101$ & $1.702$\\
VBF [\%]     & $0.026$ & $0.979$ & $0.023$\\
\bottomrule
\end{tabular}\par}
\caption{Selection efficiencies for the different Higgs production mechanisms in
  the $H\to\tau\tau$ event categories defined by CMS \cite{CMS-PAS-HIG-12-043}
  (see also Fig.~\ref{fig:signalstrengths}). The columns stand for gluon fusion
  (ggF), vector boson fusion (VBF) and $WH$ or $ZH$ associated production (VH).}
\label{tab:tautaueff}
\end{table}
%
%
%
\begin{table}[p!]
{\centering\small
\begin{tabular}{lllllll}
\toprule
& & ggF & VBF & WH & ZH & ttH \\
\midrule
ATLAS 7 TeV & ucl & $0.929$ & $0.040$ & $0.018$ & $0.010$ & $0.002$\\
& uch & $0.665$ & $0.157$ & $0.099$ & $0.057$ & $0.024$\\
& url & $0.928$ & $0.039$ & $0.020$ & $0.011$ & $0.002$\\
& urh & $0.654$ & $0.161$ & $0.108$ & $0.061$ & $0.018$\\
& ccl & $0.928$ & $0.040$ & $0.019$ & $0.010$ & $0.002$\\
& cch & $0.666$ & $0.153$ & $0.100$ & $0.057$ & $0.025$\\
& crl & $0.928$ & $0.038$ & $0.020$ & $0.011$ & $0.002$\\
& crh & $0.653$ & $0.160$ & $0.110$ & $0.059$ & $0.018$\\
& ct & $0.894$ & $0.052$ & $0.033$ & $0.017$ & $0.003$\\
& jj & $0.225$ & $0.767$ & $0.004$ & $0.002$ & $0.001$\\
\midrule
ATLAS 8 TeV & ucl & $0.937$ & $0.040$ & $0.014$ & $0.008$ & $0.002$\\
& uch & $0.793$ & $0.126$ & $0.041$ & $0.025$ & $0.014$\\
& url & $0.932$ & $0.040$ & $0.016$ & $0.010$ & $0.001$\\
& urh & $0.781$ & $0.133$ & $0.047$ & $0.028$ & $0.011$\\
& ccl & $0.936$ & $0.040$ & $0.013$ & $0.009$ & $0.002$\\
& cch & $0.789$ & $0.126$ & $0.043$ & $0.027$ & $0.015$\\
& crl & $0.932$ & $0.041$ & $0.016$ & $0.010$ & $0.001$\\
& crh & $0.777$ & $0.130$ & $0.052$ & $0.030$ & $0.011$\\
& ct & $0.907$ & $0.055$ & $0.022$ & $0.013$ & $0.002$\\
& lhm2j & $0.450$ & $0.541$ & $0.005$ & $0.003$ & $0.001$\\
& thm2j & $0.238$ & $0.760$ & $0.001$ & $0.001$ & $0.000$\\
& lm2j & $0.481$ & $0.030$ & $0.297$ & $0.172$ & $0.019$\\
& etmiss & $0.041$ & $0.005$ & $0.357$ & $0.476$ & $0.121$\\
& lept & $0.022$ & $0.006$ & $0.632$ & $0.154$ & $0.186$\\
\midrule
CMS 7 TeV & u0 & $0.614$ & $0.168$ & $0.121$ & $0.066$ & $0.031$\\
& u1 & $0.876$ & $0.062$ & $0.036$ & $0.020$ & $0.005$\\
& u2 & $0.913$ & $0.044$ & $0.025$ & $0.014$ & $0.003$\\
& u3 & $0.913$ & $0.044$ & $0.026$ & $0.015$ & $0.002$\\
& jj & $0.268$ & $0.725$ & $0.004$ & $0.002$ & $0.000$\\
\midrule
CMS 8 TeV & u0 & $0.729$ & $0.116$ & $0.082$ & $0.047$ & $0.026$\\
& u1 & $0.835$ & $0.084$ & $0.045$ & $0.026$ & $0.010$\\
& u2 & $0.916$ & $0.045$ & $0.023$ & $0.013$ & $0.004$\\
& u3 & $0.925$ & $0.039$ & $0.021$ & $0.012$ & $0.003$\\
& jj t & $0.207$ & $0.789$ & $0.002$ & $0.001$ & $0.001$\\
& jj l & $0.470$ & $0.509$ & $0.011$ & $0.006$ & $0.005$\\
& mu & $0.000$ & $0.002$ & $0.504$ & $0.286$ & $0.208$\\
& e & $0.011$ & $0.004$ & $0.502$ & $0.285$ & $0.198$\\
& Etmiss & $0.220$ & $0.026$ & $0.407$ & $0.230$ & $0.117$\\
\bottomrule
\end{tabular}\par}
\caption{Fractional contributions of the different Higgs production mechanisms
  to the $H\to\gamma\gamma$ event categories defined by ATLAS and CMS. The
  numbers and category labels are from
  \cite{ATLAS-CONF-2012-091a,ATLAS-CONF-2013-012,CMS-PAS-HIG-13-001} (see also
  Fig.~\ref{fig:signalstrengths}).  The columns stand for gluon fusion (ggF),
  vector boson fusion (VBF), $WH$ associated production (WH), $ZH$ associated
  production (ZH) and $t\bar tH$ associated production (ttH).}
\label{tab:pcvalues}
\end{table}
%
%
%
\begin{figure}[p!]
\centering
\setlength{\unitlength}{0.94\unitlength}
\begin{picture}(415,635)(-65,0)
    \SetScale{0.94}
    \SetWidth{0.2}
    \SetColor{Gray}	%	160=-4, 200=0, 300=10
    \Line(140,4)(140,633)
    \Line(160,4)(160,633)
    \Line(180,4)(180,633)
    \Line(200,4)(200,633)
    \Line(220,4)(220,633)
    \Line(240,4)(240,633)
    \Line(260,4)(260,633)
    \Line(280,4)(280,633)
    \Line(300,4)(300,633)
    \Line(320,4)(320,633)
    \Line(340,4)(340,633)
    \SetWidth{0.1}
    \SetColor{Green}
    \CBox(209.09,7)(211.06,633){Green}{Green}
    \SetWidth{1.1}
    \SetColor{Black}
    \EBox(135,7)(350,633)
    \SetWidth{2.5}
    \SetColor{Blue}
    \Vertex(215.90,630){2}
    \Line(222.80,630)(208.70,630) % {"ssTb",1.59,0.69,-0.72},215.90	222.80	208.70
    \Vertex(259.70,620){2}
    \Line(293.60,620)(228.50,620) % {"ssTg",5.97,3.39,-3.12},259.70	293.60	228.50
    \Vertex(216.80,610){2}
    \Line(239.60,610)(200.00,610) % {"ssTt",1.68,2.28,-1.68},216.80	239.60	200.00
    \Vertex(209.40,600){2}
    \Line(217.90,600)(201.10,600) % {"ssTW",0.94,0.85,-0.83},209.40	217.90	201.10
    \Vertex(204.81,590){2}
    \Line(226.66,590)(182.96,590) % {"ssAsb",0.481,2.185},204.81	226.66	182.96
    \Vertex(228.96,580){2}
    \Line(247.83,580)(210.09,580) % {"ssAsg2j",2.896,1.887},228.96	247.83	210.09
    \Vertex(163.35,570){2}
    \Line(182.51,570)(144.19,570) % {"ssAsgcch",-3.665,1.916},163.35	182.51	144.19
    \Vertex(252.76,560){2}
    \Line(278.30,560)(227.22,560) % {"ssAsgccl",5.276,2.554},252.76	278.30	227.22
    \Vertex(193.25,550){2}
    \Line(221.70,550)(165.09,550) % {"ssAsgcrh",-0.675,2.845,-2.816},193.25	221.70	165.09
    \Vertex(227.79,540){2}
    \Line(247.24,540)(208.63,540) % {"ssAsgcrl",2.779,1.945,-1.916},227.79	247.24	208.63
    \Vertex(203.70,530){2}
    \Line(239.98,530)(167.71,530) % {"ssAsgct",0.370,3.628,-3.599},203.70	239.98	167.71
    \Vertex(200.22,520){2}
    \Line(219.09,520)(181.06,520) % {"ssAsguch",0.022,1.887,-1.916},200.22	219.09	181.06
    \Vertex(206.02,510){2}
    \Line(220.24,510)(191.80,510) % {"ssAsgucl",0.602,1.422},206.02	220.24	191.80
    \Vertex(309.36,500){2}
    \Line(345.93,500)(272.79,500) % {"ssAsgurh",10.936,3.657},309.36	345.93	272.79
    \Vertex(219.67,490){2}
    \Line(235.64,490)(203.70,490) % {"ssAsgurl",1.967,1.597},219.67	235.64	203.70
    \Vertex(204.07,480){2}
    \Line(220.37,480)(183.70,480) % {"ssAst",0.407,1.630,-2.037},204.07	220.37	183.70
    \Vertex(200.00,470){2}
    \Line(206.00,470)(194.00,470) % {"ssAsW",0.0,0.6,-0.6},200.00	206.00	194.00
    \Vertex(211.85,460){2}
    \Line(224.07,460)(203.70,460) % {"ssAsZ",1.185,1.222,-0.815},211.85	224.07	203.70
    \Vertex(210.00,450){2}
    \Line(230.00,450)(190.00,450) % {"ssAeb",1.0,0.9,1.1},210.00	230.00	190.00
    \Vertex(227.11,440){2}
    \Line(246.91,440)(210.54,440) % {"ssAeglept",2.711,1.980,-1.657},227.11	246.91	210.54
    \Vertex(220.05,430){2}
    \Line(235.24,430)(207.47,430) % {"ssAegcch",2.005,1.519,-1.258},220.05	235.24	207.47
    \Vertex(213.91,420){2}
    \Line(224.34,420)(204.40,420) % {"ssAegccl",1.391,1.043,-0.951},213.91	224.34	204.40
    \Vertex(212.99,410){2}
    \Line(226.19,410)(200.25,410) % {"ssAegcrh",1.299,1.320,-1.274},212.99	226.19	200.25
    \Vertex(222.20,400){2}
    \Line(233.86,400)(212.23,400) % {"ssAegcrl",2.220,1.166,-0.997},222.20	233.86	212.23
    \Vertex(228.18,390){2}
    \Line(245.06,390)(212.22,390) % {"ssAegct",2.818,1.688,-1.596},228.18	245.06	212.22
    \Vertex(227.72,380){2}
    \Line(245.52,380)(213.91,380) % {"ssAeglhm2j",2.772,1.780,-1.381},227.72	245.52	213.91
    \Vertex(203.32,370){2}
    \Line(220.66,370)(188.74,370) % {"ssAeglm2j",0.332,1.734,-1.458},203.32	220.66	188.74
    \Vertex(229.87,360){2}
    \Line(257.03,360)(208.23,360) % {"ssAegET",2.987,2.716,-2.164},229.87	257.03	208.23
    \Vertex(216.21,350){2}
    \Line(224.50,350)(209.46,350) % {"ssAegthm2j",1.621,0.829,-0.675},216.21	224.50	209.46
    \Vertex(209.62,340){2}
    \Line(220.52,340)(200.26,340) % {"ssAeguch",0.962,1.090,-0.936},209.62	220.52	200.26
    \Vertex(208.85,330){2}
    \Line(216.06,330)(201.79,330) % {"ssAegucl",0.885,0.721,-0.706},208.85	216.06	201.79
    \Vertex(227.11,320){2}
    \Line(240.61,320)(215.60,320) % {"ssAegurh",2.711,1.350,-1.151},227.11	240.61	215.60
    \Vertex(225.27,310){2}
    \Line(234.48,310)(217.44,310) % {"ssAegurl",2.527,0.921,-0.783},225.27	234.48	217.44
    \Vertex(207.56,300){2}
    \Line(215.31,300)(200.11,300) % {"ssAet",0.756,0.775,-0.745},207.56	215.31	200.11
    \Vertex(212.60,290){2}
    \Line(216.10,290)(209.10,290) % {"ssAeW",1.26,0.35},212.60	216.10	209.10
    \Vertex(216.03,280){2}
    \Line(220.26,280)(211.44,280) % {"ssAeZ",1.603,0.423,-0.459},216.03	220.26	211.44
    \Vertex(192.71,270){2}
    \Line(212.89,270)(174.18,270) % {"ssCsbtt",-0.729,2.018,-1.853},192.71	212.89	174.18
    \Vertex(205.88,260){2}
    \Line(218.23,260)(194.35,260) % {"ssCsb",0.588,1.235,-1.153},205.88	218.23	194.35
    \Vertex(238.32,250){2}
    \Line(258.74,250)(221.61,250) % {"ssCsg0",3.832,2.042,-1.671},238.32	258.74	221.61
    \Vertex(201.93,240){2}
    \Line(211.95,240)(192.28,240) % {"ssCsg1",0.193,1.002,-0.965},201.93	211.95	192.28
    \Vertex(200.45,230){2}
    \Line(213.07,230)(187.83,230) % {"ssCsg2",0.045,1.262},200.45	213.07	187.83
    \Vertex(214.93,220){2}
    \Line(231.27,220)(198.59,220) % {"ssCsg3",1.493,1.634},214.93	231.27	198.59
    \Vertex(242.03,210){2}
    \Line(265.42,210)(224.21,210) % {"ssCsgjj",4.203,2.339,-1.782},242.03	265.42	224.21
    \Vertex(210.00,200){2}
    \Line(224.41,200)(196.00,200) % {"ssCstjet",1.000,1.441,-1.400},210.00	224.41	196.00
    \Vertex(182.82,190){2}
    \Line(196.00,190)(171.29,190) % {"ssCstVBF",-1.718,1.318,-1.153},182.82	196.00	171.29
    \Vertex(206.71,180){2}
    \Line(247.47,180)(176.24,180) % {"ssCstVH",0.671,4.076,-3.047},206.71	247.47	176.24
    \Vertex(207.26,170){2}
    \Line(211.43,170)(203.14,170) % {"ssCsW",0.726,0.417,-0.412},207.26	211.43	203.14
    \Vertex(206.71,160){2}
    \Line(213.71,160)(201.77,160) % {"ssCsZ",0.671,0.700,-0.494},206.71	213.71	201.77
    \Vertex(215.84,150){2}
    \Line(223.55,150)(208.80,150) % {"ssCeb",1.584,0.771,-0.704},215.84	223.55	208.80
    \Vertex(221.98,140){2}
    \Line(231.26,140)(213.81,140) % {"ssCeg0",2.198,0.928,-0.817},221.98	231.26	213.81
    \Vertex(200.45,130){2}
    \Line(207.50,130)(193.40,130) % {"ssCeg1",0.045,0.705},200.45	207.50	193.40
    \Vertex(203.04,120){2}
    \Line(207.87,120)(198.21,120) % {"ssCeg2",0.304,0.483},203.04	207.87	198.21
    \Vertex(196.73,110){2}
    \Line(204.90,110)(188.19,110) % {"ssCeg3",-0.327,0.817,-0.854},196.73	204.90	188.19
    \Vertex(193.39,100){2}
    \Line(221.24,100)(173.71,100) % {"ssCege",-0.661,2.785,-1.968},193.39	221.24	173.71
    \Vertex(208.24,90){2}
    \Line(219.01,90)(198.22,90) % {"ssCegjjl",0.824,1.077,-1.002},208.24	219.01	198.22
    \Vertex(203.04,80){2}
    \Line(209.72,80)(197.10,80) % {"ssCegjjt",0.304,0.668,-0.594},203.04	209.72	197.10
    \Vertex(219.38,70){2}
    \Line(245.37,70)(195.99,70) % {"ssCegET",1.938,2.599,-2.339},219.38	245.37	195.99
    \Vertex(204.16,60){2}
    \Line(222.35,60)(190.05,60) % {"ssCegm",0.416,1.819,-1.411},204.16	222.35	190.05
    \Vertex(211.09,50){2}
    \Line(215.25,50)(206.92,50) % {"ssCetjet",1.109,0.416,-0.417},211.09	215.25	206.92
    \Vertex(223.75,40){2}
    \Line(230.49,40)(216.72,40) % {"ssCetVBF",2.375,0.674,-0.703},223.75	230.49	216.72
    \Vertex(207.80,30){2}
    \Line(223.93,30)(190.57,30) % {"ssCetVH",0.780,1.613,-1.723},207.80	223.93	190.57
    \Vertex(206.47,20){2}
    \Line(208.72,20)(204.26,20) % {"ssCeW",0.647,0.225,-0.221},206.47	208.72	204.26
    \Vertex(209.75,10){2}
    \Line(213.07,10)(207.00,10) % {"ssCeZ",0.975,0.332,-0.275}209.75	213.07	207.00
    \SetWidth{1.0}
    \SetColor{Black}
% {"ssTb",1.585,0.710,-0.747},
    \Text(-65,630)[cl]{\Black{\scriptsize $\mu ^{\rm T}(VH\to Vb\bar b)=1.59 ^{+0.69} {}_{-0.72}$}}
    \Text(100,630)[cl]{\Black{\scriptsize \cite{CDF:2013kxa}}}
% {"ssTg",6.138,3.260,-3.188},
    \Text(-65,620)[cl]{\Black{\scriptsize $\mu ^{\rm T}(H\to \gamma \gamma )=5.97 ^{+3.39} {}_{-3.12}$}}
    \Text(100,620)[cl]{\Black{\scriptsize \cite{CDF:2013kxa}}}
% {"ssTt",2.077,2.168,-1.949},
    \Text(-65,610)[cl]{\Black{\scriptsize $\mu ^{\rm T}(H\to \tau \tau )=1.68 ^{+2.28} {}_{-1.68}$}}
    \Text(100,610)[cl]{\Black{\scriptsize \cite{CDF:2013kxa}}}
% {"ssTW",0.856,0.893,-0.801},
    \Text(-65,600)[cl]{\Black{\scriptsize $\mu ^{\rm T}(H\to WW)=0.94 ^{+0.85} {}_{-0.83}$}}
    \Text(100,600)[cl]{\Black{\scriptsize \cite{CDF:2013kxa}}}
% {"ssAsb",0.481,2.185},
    \Text(-65,590)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(VH\to Vb\bar b)=0.481 \pm 2.185$}}
    \Text(100,590)[cl]{\Black{\scriptsize \cite{Aad:2012an}}}
% {"ssAsg2j",2.896,1.887},
    \Text(-65,580)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny jj}}=2.896 \pm 1.887$}}
    \Text(100,580)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgcch",-3.665,1.916},
    \Text(-65,570)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny cch}}=-3.665 \pm 1.916$}}
    \Text(100,570)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgccl",5.276,2.554},
    \Text(-65,560)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny ccl}}=5.276 \pm 2.554$}}
    \Text(100,560)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgcrh",-0.675,2.845,-2.816},
    \Text(-65,550)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny crh}}=-0.675 ^{+2.845} {}_{-2.816}$}}
    \Text(100,550)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgcrl",2.779,1.945,-1.916},
    \Text(-65,540)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny crl}}=2.779 ^{+1.945} {}_{-1.916}$}}
    \Text(100,540)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgct",0.370,3.628,-3.599},
    \Text(-65,530)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny ct}}=0.370 ^{+3.628} {}_{-3.599}$}}
    \Text(100,530)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsguch",0.022,1.887,-1.916},
    \Text(-65,520)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny uch}}=0.022 ^{+1.887} {}_{-1.916}$}}
    \Text(100,520)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgucl",0.602,1.422},
    \Text(-65,510)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny ucl}}=0.602 \pm 1.422$}}
    \Text(100,510)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgurh",10.936,3.657},
    \Text(-65,500)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny urh}}=10.936 \pm 3.657$}}
    \Text(100,500)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAsgurl",1.967,1.597},
    \Text(-65,490)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \gamma \gamma )_{\text{\tiny url}}=1.967 \pm 1.597$}}
    \Text(100,490)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-091}}}
% {"ssAst",0.407,1.630,-2.037},
    \Text(-65,480)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to \tau \tau )=0.407 ^{+1.630} {}_{-2.037}$}}
    \Text(100,480)[cl]{\Black{\scriptsize \cite{Aad:2012an}}}
% {"ssAsW",0.0,0.6,-0.6},
    \Text(-65,470)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to WW)=0.0 ^{+0.6} {}_{-0.6}$}}
    \Text(100,470)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-030}}}
% {"ssAsZ",1.185,1.222,-0.815},
    \Text(-65,460)[cl]{\Black{\scriptsize $\mu ^{\rm A7}(H\to ZZ)=1.185 ^{+1.222} {}_{-0.815}$}}
    \Text(100,460)[cl]{\Black{\scriptsize \cite{Aad:2012an}}}
% {"ssAeb",1.0,0.9,1.1},
    \Text(-65,450)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(VH\to Vb\bar b)=1.0 \pm 0.9 \pm 1.1$}}
    \Text(100,450)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2012-161}}}
% {"ssAeglept",2.711,1.980,-1.657},
    \Text(-65,440)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny lept}}=2.711 ^{+1.980} {}_{-1.657}$}}
    \Text(100,440)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegcch",2.005,1.519,-1.258},
    \Text(-65,430)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny cch}}=2.005 ^{+1.519} {}_{-1.258}$}}
    \Text(100,430)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegccl",1.391,1.043,-0.951},
    \Text(-65,420)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny ccl}}=1.391 ^{+1.043} {}_{-0.951}$}}
    \Text(100,420)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegcrh",1.299,1.320,-1.274},
    \Text(-65,410)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny crh}}=1.299 ^{+1.320} {}_{-1.274}$}}
    \Text(100,410)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegcrl",2.220,1.166,-0.997},
    \Text(-65,400)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny crl}}=2.220 ^{+1.166} {}_{-0.997}$}}
    \Text(100,400)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegct",2.818,1.688,-1.596},
    \Text(-65,390)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny ct}}=2.818 ^{+1.688} {}_{-1.596}$}}
    \Text(100,390)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAeglhm2j",2.772,1.780,-1.381},
    \Text(-65,380)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny lhmjj}}=2.772 ^{+1.780} {}_{-1.381}$}}
    \Text(100,380)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAeglm2j",0.332,1.734,-1.458},
    \Text(-65,370)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny lmjj}}=0.332 ^{+1.734} {}_{-1.458}$}}
    \Text(100,370)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegET",2.987,2.716,-2.164},
    \Text(-65,360)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny $\fmslash{E}_{\rm T}$}}=2.987 ^{+2.716} {}_{-2.164}$}}
    \Text(100,360)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegthm2j",1.621,0.829,-0.675},
    \Text(-65,350)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny thmjj}}=1.621 ^{+0.829} {}_{-0.675}$}}
    \Text(100,350)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAeguch",0.962,1.090,-0.936},
    \Text(-65,340)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny uch}}=0.962 ^{+1.090} {}_{-0.936}$}}
    \Text(100,340)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegucl",0.885,0.721,-0.706},
    \Text(-65,330)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny ucl}}=0.885 ^{+0.721} {}_{-0.706}$}}
    \Text(100,330)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegurh",2.711,1.350,-1.151},
    \Text(-65,320)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny urh}}=2.711 ^{+1.350} {}_{-1.151}$}}
    \Text(100,320)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAegurl",2.527,0.921,-0.783},
    \Text(-65,310)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \gamma \gamma )_{\text{\tiny url}}=2.527 ^{+0.921} {}_{-0.783}$}}
    \Text(100,310)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-012}}}
% {"ssAet",0.756,0.775,-0.745},
    \Text(-65,300)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to \tau \tau )=0.756 ^{+0.775} {}_{-0.745}$}}
    \Text(100,300)[cl]{\Black{\scriptsize \cite{Aad:2012an,ATLAS-CONF-2012-160}}}
% {"ssAeW",1.26,0.35},
    \Text(-65,290)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to WW)=1.26 \pm 0.35$}}
    \Text(100,290)[cl]{\Black{\scriptsize \cite{ATLAS-CONF-2013-030}}}
% {"ssAeZ",1.603,0.423,-0.459},
    \Text(-65,280)[cl]{\Black{\scriptsize $\mu ^{\rm A8}(H\to ZZ)=1.603 ^{+0.423} {}_{-0.459}$}}
    \Text(100,280)[cl]{\Black{\scriptsize \cite{Aad:2012an,ATLAS-CONF-2013-013}}}
% {"ssCsbtt",-0.729,2.018,-1.853},
    \Text(-65,270)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(t\bar t\to H\to b\bar b)=-0.729 ^{+2.018} {}_{-1.853}$}}
    \Text(100,270)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020}}}
% {"ssCsb",0.588,1.235,-1.153},
    \Text(-65,260)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(VH\to Vb\bar b)=0.588 ^{+1.235} {}_{-1.153}$}}
    \Text(100,260)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020}}}
% {"ssCsg0",3.832,2.042,-1.671},
    \Text(-65,250)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \gamma \gamma )_{\text{\tiny u0}}=3.832 ^{+2.042} {}_{-1.671}$}}
    \Text(100,250)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCsg1",0.193,1.002,-0.965},
    \Text(-65,240)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \gamma \gamma )_{\text{\tiny u1}}=0.193 ^{+1.002} {}_{-0.965}$}}
    \Text(100,240)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCsg2",0.045,1.262},
    \Text(-65,230)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \gamma \gamma )_{\text{\tiny u2}}=0.045 \pm 1.262$}}
    \Text(100,230)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCsg3",1.493,1.634},
    \Text(-65,220)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \gamma \gamma )_{\text{\tiny u3}}=1.493 \pm 1.634$}}
    \Text(100,220)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCsgjj",4.203,2.339,-1.782},
    \Text(-65,210)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \gamma \gamma )_{\text{\tiny jj}}=4.203 ^{+2.339} {}_{-1.782}$}}
    \Text(100,210)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCstjet",1.000,1.441,-1.400},
    \Text(-65,200)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \tau \tau )_{\text{\tiny 0/1 jet}}=1.000 ^{+1.441} {}_{-1.400}$}}
    \Text(100,200)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020}}}
% {"ssCstVBF",-1.718,1.318,-1.153},
    \Text(-65,190)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \tau \tau )_{\text{\tiny VBF}}=-1.718 ^{+1.318} {}_{-1.153}$}}
    \Text(100,190)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020}}}
% {"ssCstVH",0.671,4.076,-3.047},
    \Text(-65,180)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to \tau \tau )_{\text{\tiny VH}}=0.671 ^{+4.076} {}_{-3.047}$}}
    \Text(100,180)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020}}}
% {"ssCsW",0.726,0.417,-0.412},
    \Text(-65,170)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to WW)=0.726 ^{+0.417} {}_{-0.412}$}}
    \Text(100,170)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-003}}}
% {"ssCsZ",0.671,0.700,-0.494},
    \Text(-65,160)[cl]{\Black{\scriptsize $\mu ^{\rm C7}(H\to ZZ)=0.671 ^{+0.700} {}_{-0.494}$}}
    \Text(100,160)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020}}}
% {"ssCeb",1.584,0.771,-0.704},
    \Text(-65,150)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(VH\to Vb\bar b)=1.584 ^{+0.771} {}_{-0.704}$}}
    \Text(100,150)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020,CMS-PAS-HIG-12-045}}}
% {"ssCeg0",2.198,0.928,-0.817},
    \Text(-65,140)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny u0}}=2.198 ^{+0.928} {}_{-0.817}$}}
    \Text(100,140)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCeg1",0.045,0.705},
    \Text(-65,130)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny u1}}=0.045 \pm 0.705$}}
    \Text(100,130)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCeg2",0.304,0.483},
    \Text(-65,120)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny u2}}=0.304 \pm 0.483$}}
    \Text(100,120)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCeg3",-0.327,0.817,-0.854},
    \Text(-65,110)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny u3}}=-0.327 ^{+0.817} {}_{-0.854}$}}
    \Text(100,110)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCege",-0.661,2.785,-1.968},
    \Text(-65,100)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny $e$}}=-0.661 ^{+2.785} {}_{-1.968}$}}
    \Text(100,100)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCegjjl",0.824,1.077,-1.002},
    \Text(-65,90)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny jj loose}}=0.824 ^{+1.077} {}_{-1.002}$}}
    \Text(100,90)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCegjjt",0.304,0.668,-0.594},
    \Text(-65,80)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny jj tight}}=0.304 ^{+0.668} {}_{-0.594}$}}
    \Text(100,80)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCegET",1.938,2.599,-2.339},
    \Text(-65,70)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny $\fmslash{E}_{\rm T}$}}=1.938 ^{+2.599} {}_{-2.339}$}}
    \Text(100,70)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCegm",0.416,1.819,-1.411},
    \Text(-65,60)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \gamma \gamma )_{\text{\tiny $\mu$}}=0.416 ^{+1.819} {}_{-1.411}$}}
    \Text(100,60)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-001}}}
% {"ssCetjet",1.109,0.416,-0.417},
    \Text(-65,50)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \tau \tau )_{\text{\tiny 0/1 jet}}=1.109 ^{+0.416} {}_{-0.417}$}}
    \Text(100,50)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020,CMS-PAS-HIG-13-004}}}
% {"ssCetVBF",2.375,0.674,-0.703},
    \Text(-65,40)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \tau \tau )_{\text{\tiny VBF}}=2.375 ^{+0.674} {}_{-0.703}$}}
    \Text(100,40)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020,CMS-PAS-HIG-13-004}}}
% {"ssCetVH",0.780,1.613,-1.723},
    \Text(-65,30)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to \tau \tau )_{\text{\tiny VH}}=0.780 ^{+1.613} {}_{-1.723}$}}
    \Text(100,30)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020,CMS-PAS-HIG-13-004}}}
% {"ssCeW",0.647,0.225,-0.221},
    \Text(-65,20)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to WW)=0.647 ^{+0.225} {}_{-0.221}$}}
    \Text(100,20)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-13-003}}}
% {"ssCeZ",0.975,0.332,-0.275}
    \Text(-65,10)[cl]{\Black{\scriptsize $\mu ^{\rm C8}(H\to ZZ)=0.975 ^{+0.332} {}_{-0.275}$}}
    \Text(100,10)[cl]{\Black{\scriptsize \cite{CMS-PAS-HIG-12-020,CMS-PAS-HIG-13-002}}}
    \Text(139,-3)[cb]{\Black{\scriptsize $-6$}}
    \Text(159,-3)[cb]{\Black{\scriptsize $-4$}}
    \Text(179,-3)[cb]{\Black{\scriptsize $-2$}}
    \Text(200.7,-3)[cb]{\Black{\scriptsize $0$}}
    \Text(221,-3)[cb]{\Black{\scriptsize $2$}}
    \Text(241,-3)[cb]{\Black{\scriptsize $4$}}
    \Text(261,-3)[cb]{\Black{\scriptsize $6$}}
    \Text(281,-3)[cb]{\Black{\scriptsize $8$}}
    \Text(301.7,-3)[cb]{\Black{\scriptsize $10$}}
    \Text(321.7,-3)[cb]{\Black{\scriptsize $12$}}
    \Text(341.7,-3)[cb]{\Black{\scriptsize $14$}}
%     \put(290,80){\includegraphics[width=0.13\linewidth]{Images/CKMfitterPackage.eps}}
\end{picture}
\caption{Higgs signal strengths measured by D0, CDF, ATLAS and CMS. The
  superscripts T, A7, A8, C7 and C8 refer to Tevatron, ATLAS \unit{7}{TeV},
  ATLAS \unit{8}{TeV}, CMS \unit{7}{TeV} and CMS \unit{8}{TeV} data,
  respectively. The subscripts jj, cch etc.\ denote the different event
  categories defined by ATLAS and CMS. The combination of all signal strengths
  yields $\mu _{\text{combined}}=1.007 ^{+0.099} {}_{-0.098}$ and is
  illustrated by the green band.}
\label{fig:signalstrengths}
\end{figure}

